Author's response to reviews

Title: The Primary Care Practitioner and the Diagnosis of Occupational Diseases

Authors:

Luca Cegolon (mrcegolon@hotmail.com)
John H Lange (jhlan1g@hotmail.com)
Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)

Version: 5 Date: 1 July 2010

Author's response to reviews:

Dear Editor,

I hereby submit a slightly modified version of the paper entitled: "The Primary Care Practitioner and the Diagnosis of Occupational Diseases", by Cegolon L et al.

The structure and the content of the whole manuscript have not been changed, but in my opinion the new version contains some amendments that render the paper more convincing.

I have highlighted additional points in blue and minor changes to the original are highlighted in yellow.

I hope you will agree that these amendments improve the paper and look forward to hearing your opinion.

Best Wishes.

Dr. Luca Cegolon, MD, FFPH

Corresponding Author